Dear Editor,

Thank you for the opportunity to resubmit our article "The impact of a computerized decision aid on empowering pregnant women for choosing vaginal versus cesarean section delivery - Study protocol for a randomized controlled trial" by Saeid Eslami, Azam Aslani, Fatemeh Tara, Lila Ghalichi, Fatemeh Erfanian, Ameen Abu-Hanna. Enclosed please find the re-submission for possible publication in Trials.

We have addressed the two formatting remarks. First, we changed the introduction" to “Background”. Second, we added an Acknowledgements section at the end of the manuscript before the reference list. All changes are marked in the text.

We trust that the resubmitted paper meets all points made by you and hope that it is suitable for publication in its new form.

Sincerely,